Author’s response to reviews

Title: Effectiveness of a website and mobile phone based physical activity and nutrition intervention for middle-aged males: Trial protocol and baseline findings of the ManUp Study

Authors:

Mitch J Duncan (m.duncan@cqu.edu.au)
Corneel Vandelanotte (c.vandelanotte@cqu.edu.au)
Richard R Rosenkranz (ricardo@ksu.edu)
Cristina M Caperchione (cristina.caperchione@ubc.ca)
Hang Ding (Hang.Ding@csiro.au)
Marcus Ellison (m.ellison@cqu.edu.au)
Emma S George (e.george@uws.edu.au)
Cindy Hooker (c.hooker@cqu.edu.au)
Mohan Karunanithi (Mohan.Karunanithi@csiro.au)
Gregory S Kolt (g.kolt@uws.edu.au)
Anthony Maeder (a.maeder@uws.edu.au)
Manny Noakes (Manny.Noakes@csiro.au)
Rhys Tague (r.tague@uws.edu.au)
Pennie Taylor (Pennie.Taylor@csiro.au)
Pierre Viljoen (p.viljoen@cqu.edu.au)
W Kerry Mummery (kerry.mummery@ualberta.ca)

Version: 2 Date: 27 June 2012

Author’s response to reviews: see over